\documentclass[kind=ENG,a4j,graduate,landscape,viewer,color,check]{edb-syl}
\def\年度{2011}
\def\和年度{23}
\begin{document}

\Englishfalse
%%%% 216534:216592
\begin{edb:lecture}%
\EdbEID{216592}\EdbEOID{554709}%
\JabeeEfalse%
\JabeeRfalse%
\包含false%
\授業ID{216534:216592}%
\授業概要種別{166222}%
\名称{加工システム}{Production and Manufacturing System}%
\担当教員{\所属肩書氏名{知的力学システム工学専攻 機械創造システム工学コース 生産システム講座}{准教授}{\href{http://pub2.db.tokushima-u.ac.jp/ERD/person/10659/profile-ja.html}{\EdbSMG{多田}{}{吉宏}}}}%
\開講学期{}%
\履修種別{選択}%
\EdbColDef{edb:lecture.goal}{1}{E}{To understand the basic concepts and related aspects of work measurement and predetermined time standard system.}%
\EdbColDef{edb:lecture.goal}{1}{J}{作業測定や作業標準時間設定の基本的な考え方を理解し分析に適用できること}%
\EdbColDef{edb:lecture.plan}{1}{E}{Introduction to work analysis}%
\EdbColDef{edb:lecture.plan}{1}{J}{作業分析の基礎}%
\EdbColDef{edb:lecture.plan}{2}{E}{Work study and time study}%
\EdbColDef{edb:lecture.plan}{2}{J}{動作分析・時間分析}%
\EdbColDef{edb:lecture.plan}{3}{E}{Rating techniques}%
\EdbColDef{edb:lecture.plan}{3}{J}{各種のレイティング手法}%
\EdbColDef{edb:lecture.plan}{4}{E}{Predetermined time standard system}%
\EdbColDef{edb:lecture.plan}{4}{J}{既定時間標準法}%
\EdbColDef{edb:lecture.plan}{5}{E}{Fundamentals of MOST}%
\EdbColDef{edb:lecture.plan}{5}{J}{basic MOSTの基礎}%
\EdbColDef{edb:lecture.plan}{6}{E}{General move sequence}%
\EdbColDef{edb:lecture.plan}{6}{J}{普通移動シーケンス}%
\EdbColDef{edb:lecture.plan}{7}{E}{Exercise of the general move sequence}%
\EdbColDef{edb:lecture.plan}{7}{J}{普通移動の分析と演習}%
\EdbColDef{edb:lecture.plan}{8}{E}{Controlled move sequence}%
\EdbColDef{edb:lecture.plan}{8}{J}{制限移動シーケンス}%
\EdbColDef{edb:lecture.plan}{9}{E}{Exercise of the controlled move sequence}%
\EdbColDef{edb:lecture.plan}{9}{J}{制限移動の分析と演習}%
\EdbColDef{edb:lecture.plan}{10}{E}{Analysis of continuous move}%
\EdbColDef{edb:lecture.plan}{10}{J}{連続動作の分析}%
\EdbColDef{edb:lecture.plan}{11}{E}{Exercise of continuous move}%
\EdbColDef{edb:lecture.plan}{11}{J}{連続動作の分析と演習}%
\EdbColDef{edb:lecture.plan}{12}{E}{Tool use sequence}%
\EdbColDef{edb:lecture.plan}{12}{J}{工具使用シーケンス}%
\EdbColDef{edb:lecture.plan}{13}{E}{Exercise of the tool use sequence}%
\EdbColDef{edb:lecture.plan}{13}{J}{工具使用の分析と演習}%
\EdbColDef{edb:lecture.plan}{14}{E}{Manual crane sequence}%
\EdbColDef{edb:lecture.plan}{14}{J}{手動クレーンシーケンス}%
\EdbColDef{edb:lecture.plan}{15}{E}{Exercise of the manual crane sequence}%
\EdbColDef{edb:lecture.plan}{15}{J}{手動クレーンの分析と演習}%
\EdbColDef{edb:lecture.plan}{16}{E}{Examination}%
\EdbColDef{edb:lecture.plan}{16}{J}{定期試験}%
\EdbColDef{edb:lecture.style}{1}{@}{60007}%
\EdbColDef{edb:lecture.style}{1}{E}{Lecture}%
\EdbColDef{edb:lecture.style}{1}{J}{講義}%
\EdbColDef{edb:lecture.unit}{1}{E}{2}%
\EdbColDef{edb:lecture.unit}{1}{J}{2}%
\EdbColDef{edb:lecture.target}{1}{E}{Work measurement based on predetermined time standard system for advanced manufacturing and production are introduced.}%
\EdbColDef{edb:lecture.target}{1}{J}{生産加工や組立ての高度化・高能率化について，既定時間標準法に基づいた作業測定の考え方と手法を習得する．}%
\EdbColDef{edb:lecture.outline}{1}{E}{This class introduces basic-MOST for work measurement and determination of the standard time based on predetermined time standard system.}%
\EdbColDef{edb:lecture.outline}{1}{J}{「もの作り」の効率化に有効なIE手法について，basic-MOSTを中心に作業測定および作業標準時間設定の基本的な考え方とその適用方法を講義する．}%
\EdbColDef{edb:lecture.keyword}{1}{E}{industrial engineering}%
\EdbColDef{edb:lecture.keyword}{1}{J}{インダストリアルエンジニアリング}%
\EdbColDef{edb:lecture.keyword}{2}{E}{work measurement}%
\EdbColDef{edb:lecture.keyword}{2}{J}{作業測定}%
\EdbColDef{edb:lecture.keyword}{3}{E}{standard time}%
\EdbColDef{edb:lecture.keyword}{3}{J}{作業標準時間}%
\EdbColDef{edb:lecture.relation}{1}{@}{169351}%
\EdbColDef{edb:lecture.relation}{1}{E}{\EdbSeeLecture{Theory of Plasticity and Application to Metal Forming Processes}{168446}{169351}{0.5}}%
\EdbColDef{edb:lecture.relation}{1}{J}{\EdbSeeLecture{金属加工学}{168446}{169351}{0.5}}%
\EdbColDef{edb:lecture.requirement}{1}{E}{Students are required to have a good understanding on fundamentals of material processing technologies and manufacturing system.}%
\EdbColDef{edb:lecture.requirement}{1}{J}{生産加工に関する基礎的な素養があること．}%
\EdbColDef{edb:lecture.notice}{1}{E}{Students are required to prepare and review the class}%
\EdbColDef{edb:lecture.notice}{1}{J}{予習・復習が重要である．}%
\EdbColDef{edb:lecture.evaluation}{1}{E}{Exercises and assignments count 50\%,{\EdbZSp} and examination counts 50\%.}%
\EdbColDef{edb:lecture.evaluation}{1}{J}{レポート・演習 (50 点) と試験(50 点) を総合して評価する．}%
\EdbColDef{edb:lecture.evaluation2}{1}{E}{None}%
\EdbColDef{edb:lecture.evaluation2}{1}{J}{無し}%
\EdbColDef{edb:lecture.textbook}{1}{E}{Printed synopsises are used.}%
\EdbColDef{edb:lecture.textbook}{1}{J}{プリント資料で講義する．}%
\EdbColDef{edb:lecture.reference}{1}{E}{Work study,{\EdbZSp} ISBN4-526-00084-1.}%
\EdbColDef{edb:lecture.reference}{1}{J}{通産省産業構造審議会編「作業研究」，日刊工業新聞社，ISBN4-526-00084-1.}%
\EdbColDef{edb:lecture.note}{1}{E}{Students are required preparation (2 hours) and review (2 hours) for every two hours class work.}%
\EdbColDef{edb:lecture.note}{1}{J}{授業を受ける際には，2時間の授業時間毎に2時間の予習と2時間の復習を行うことが，授業の理解と単位取得のために必要である}%
\EdbColDef{edb:lecture.contact}{1}{J}{\連絡先個人{\EdbEMAIL{tada@me.tokushima-u.ac.jp}{t{\HZSp[1]}a{\HZSp[1]}d{\HZSp[1]}a{\HZSp[1]}@{\HZSp[1]}m{\HZSp[1]}e{\HZSp[1]}.{\HZSp[1]}t{\HZSp[1]}o{\HZSp[1]}k{\HZSp[1]}u{\HZSp[1]}s{\HZSp[1]}h{\HZSp[1]}i{\HZSp[1]}m{\HZSp[1]}a{\HZSp[1]}-{\HZSp[1]}u{\HZSp[1]}.{\HZSp[1]}a{\HZSp[1]}c{\HZSp[1]}.{\HZSp[1]}j{\HZSp[1]}p}}{多田 (M319, 088-656-7381, \EdbEMAIL{tada@me.tokushima-u.ac.jp}{t{\HZSp[1]}a{\HZSp[1]}d{\HZSp[1]}a{\HZSp[1]}@{\HZSp[1]}m{\HZSp[1]}e{\HZSp[1]}.{\HZSp[1]}t{\HZSp[1]}o{\HZSp[1]}k{\HZSp[1]}u{\HZSp[1]}s{\HZSp[1]}h{\HZSp[1]}i{\HZSp[1]}m{\HZSp[1]}a{\HZSp[1]}-{\HZSp[1]}u{\HZSp[1]}.{\HZSp[1]}a{\HZSp[1]}c{\HZSp[1]}.{\HZSp[1]}j{\HZSp[1]}p})}{}}%
\end{edb:lecture}%

\Englishtrue
%%%% 216534:216592
\begin{edb:lecture}%
\EdbEID{216592}\EdbEOID{554709}%
\JabeeEfalse%
\JabeeRfalse%
\包含false%
\授業ID{216534:216592}%
\授業概要種別{166222}%
\名称{Production and Manufacturing System}{加工システム}%
\担当教員{\所属肩書氏名{Production Systems Engineering, Mechanical Engineering, Intelligent Structures and Mechanics Systems Engineering}{Associate Professor}{\href{http://pub2.db.tokushima-u.ac.jp/ERD/person/10659/profile-en.html}{\EdbGMS{Yoshihiro}{}{Tada}}}}%
\開講学期{}%
\履修種別{selection}%
\EdbColDef{edb:lecture.goal}{1}{J}{作業測定や作業標準時間設定の基本的な考え方を理解し分析に適用できること}%
\EdbColDef{edb:lecture.goal}{1}{E}{To understand the basic concepts and related aspects of work measurement and predetermined time standard system.}%
\EdbColDef{edb:lecture.plan}{1}{J}{作業分析の基礎}%
\EdbColDef{edb:lecture.plan}{1}{E}{Introduction to work analysis}%
\EdbColDef{edb:lecture.plan}{2}{J}{動作分析・時間分析}%
\EdbColDef{edb:lecture.plan}{2}{E}{Work study and time study}%
\EdbColDef{edb:lecture.plan}{3}{J}{各種のレイティング手法}%
\EdbColDef{edb:lecture.plan}{3}{E}{Rating techniques}%
\EdbColDef{edb:lecture.plan}{4}{J}{既定時間標準法}%
\EdbColDef{edb:lecture.plan}{4}{E}{Predetermined time standard system}%
\EdbColDef{edb:lecture.plan}{5}{J}{basic MOSTの基礎}%
\EdbColDef{edb:lecture.plan}{5}{E}{Fundamentals of MOST}%
\EdbColDef{edb:lecture.plan}{6}{J}{普通移動シーケンス}%
\EdbColDef{edb:lecture.plan}{6}{E}{General move sequence}%
\EdbColDef{edb:lecture.plan}{7}{J}{普通移動の分析と演習}%
\EdbColDef{edb:lecture.plan}{7}{E}{Exercise of the general move sequence}%
\EdbColDef{edb:lecture.plan}{8}{J}{制限移動シーケンス}%
\EdbColDef{edb:lecture.plan}{8}{E}{Controlled move sequence}%
\EdbColDef{edb:lecture.plan}{9}{J}{制限移動の分析と演習}%
\EdbColDef{edb:lecture.plan}{9}{E}{Exercise of the controlled move sequence}%
\EdbColDef{edb:lecture.plan}{10}{J}{連続動作の分析}%
\EdbColDef{edb:lecture.plan}{10}{E}{Analysis of continuous move}%
\EdbColDef{edb:lecture.plan}{11}{J}{連続動作の分析と演習}%
\EdbColDef{edb:lecture.plan}{11}{E}{Exercise of continuous move}%
\EdbColDef{edb:lecture.plan}{12}{J}{工具使用シーケンス}%
\EdbColDef{edb:lecture.plan}{12}{E}{Tool use sequence}%
\EdbColDef{edb:lecture.plan}{13}{J}{工具使用の分析と演習}%
\EdbColDef{edb:lecture.plan}{13}{E}{Exercise of the tool use sequence}%
\EdbColDef{edb:lecture.plan}{14}{J}{手動クレーンシーケンス}%
\EdbColDef{edb:lecture.plan}{14}{E}{Manual crane sequence}%
\EdbColDef{edb:lecture.plan}{15}{J}{手動クレーンの分析と演習}%
\EdbColDef{edb:lecture.plan}{15}{E}{Exercise of the manual crane sequence}%
\EdbColDef{edb:lecture.plan}{16}{J}{定期試験}%
\EdbColDef{edb:lecture.plan}{16}{E}{Examination}%
\EdbColDef{edb:lecture.style}{1}{@}{60007}%
\EdbColDef{edb:lecture.style}{1}{J}{講義}%
\EdbColDef{edb:lecture.style}{1}{E}{Lecture}%
\EdbColDef{edb:lecture.unit}{1}{J}{2}%
\EdbColDef{edb:lecture.unit}{1}{E}{2}%
\EdbColDef{edb:lecture.target}{1}{J}{生産加工や組立ての高度化・高能率化について，既定時間標準法に基づいた作業測定の考え方と手法を習得する．}%
\EdbColDef{edb:lecture.target}{1}{E}{Work measurement based on predetermined time standard system for advanced manufacturing and production are introduced.}%
\EdbColDef{edb:lecture.outline}{1}{J}{「もの作り」の効率化に有効なIE手法について，basic-MOSTを中心に作業測定および作業標準時間設定の基本的な考え方とその適用方法を講義する．}%
\EdbColDef{edb:lecture.outline}{1}{E}{This class introduces basic-MOST for work measurement and determination of the standard time based on predetermined time standard system.}%
\EdbColDef{edb:lecture.keyword}{1}{J}{インダストリアルエンジニアリング}%
\EdbColDef{edb:lecture.keyword}{1}{E}{industrial engineering}%
\EdbColDef{edb:lecture.keyword}{2}{J}{作業測定}%
\EdbColDef{edb:lecture.keyword}{2}{E}{work measurement}%
\EdbColDef{edb:lecture.keyword}{3}{J}{作業標準時間}%
\EdbColDef{edb:lecture.keyword}{3}{E}{standard time}%
\EdbColDef{edb:lecture.relation}{1}{@}{169351}%
\EdbColDef{edb:lecture.relation}{1}{J}{\EdbSeeLecture{金属加工学}{168446}{169351}{0.5}}%
\EdbColDef{edb:lecture.relation}{1}{E}{\EdbSeeLecture{Theory of Plasticity and Application to Metal Forming Processes}{168446}{169351}{0.5}}%
\EdbColDef{edb:lecture.requirement}{1}{J}{生産加工に関する基礎的な素養があること．}%
\EdbColDef{edb:lecture.requirement}{1}{E}{Students are required to have a good understanding on fundamentals of material processing technologies and manufacturing system.}%
\EdbColDef{edb:lecture.notice}{1}{J}{予習・復習が重要である．}%
\EdbColDef{edb:lecture.notice}{1}{E}{Students are required to prepare and review the class}%
\EdbColDef{edb:lecture.evaluation}{1}{J}{レポート・演習 (50 点) と試験(50 点) を総合して評価する．}%
\EdbColDef{edb:lecture.evaluation}{1}{E}{Exercises and assignments count 50\%,{\EdbZSp} and examination counts 50\%.}%
\EdbColDef{edb:lecture.evaluation2}{1}{J}{無し}%
\EdbColDef{edb:lecture.evaluation2}{1}{E}{None}%
\EdbColDef{edb:lecture.textbook}{1}{J}{プリント資料で講義する．}%
\EdbColDef{edb:lecture.textbook}{1}{E}{Printed synopsises are used.}%
\EdbColDef{edb:lecture.reference}{1}{J}{通産省産業構造審議会編「作業研究」，日刊工業新聞社，ISBN4-526-00084-1.}%
\EdbColDef{edb:lecture.reference}{1}{E}{Work study,{\EdbZSp} ISBN4-526-00084-1.}%
\EdbColDef{edb:lecture.note}{1}{J}{授業を受ける際には，2時間の授業時間毎に2時間の予習と2時間の復習を行うことが，授業の理解と単位取得のために必要である}%
\EdbColDef{edb:lecture.note}{1}{E}{Students are required preparation (2 hours) and review (2 hours) for every two hours class work.}%
\EdbColDef{edb:lecture.contact}{1}{E}{\連絡先個人{\EdbEMAIL{tada@me.tokushima-u.ac.jp}{t{\HZSp[1]}a{\HZSp[1]}d{\HZSp[1]}a{\HZSp[1]}@{\HZSp[1]}m{\HZSp[1]}e{\HZSp[1]}.{\HZSp[1]}t{\HZSp[1]}o{\HZSp[1]}k{\HZSp[1]}u{\HZSp[1]}s{\HZSp[1]}h{\HZSp[1]}i{\HZSp[1]}m{\HZSp[1]}a{\HZSp[1]}-{\HZSp[1]}u{\HZSp[1]}.{\HZSp[1]}a{\HZSp[1]}c{\HZSp[1]}.{\HZSp[1]}j{\HZSp[1]}p}}{Tada (M319, +81-88-656-7381, \EdbEMAIL{tada@me.tokushima-u.ac.jp}{t{\HZSp[1]}a{\HZSp[1]}d{\HZSp[1]}a{\HZSp[1]}@{\HZSp[1]}m{\HZSp[1]}e{\HZSp[1]}.{\HZSp[1]}t{\HZSp[1]}o{\HZSp[1]}k{\HZSp[1]}u{\HZSp[1]}s{\HZSp[1]}h{\HZSp[1]}i{\HZSp[1]}m{\HZSp[1]}a{\HZSp[1]}-{\HZSp[1]}u{\HZSp[1]}.{\HZSp[1]}a{\HZSp[1]}c{\HZSp[1]}.{\HZSp[1]}j{\HZSp[1]}p})}{}}%
\end{edb:lecture}%

\end{document}
\end{document}


